Author's response to reviews

Title: Extramedullary hematopoiesis presenting as a compressive cord and cerebral lesion in a patient without a significant Hematologic Disorder

Authors:

Amir Saied Seddighi (invincible19152@gmail.com)
Afsoun Seddighi (invincible19152@yahoo.com)

Version: 4 Date: 28 August 2009

Author's response to reviews: see over
Dear Editor  
Thanks a lot for your guidelines to correct the format of our article. As your order, we corrected the grammatical writing errors. We also added the complementary neuroimaging and the slide of peripheral blood smear of the patient. We added the information requested by the reviewers including the explanation that the patient had not thalassemia minor because the hemoglobin electrophoresis was normal. The cause of thrombocytopenia was mild splenomegaly of the patient. The CBC analysis in the second admission was typed incorrectly which was corrected. 
Since this is a very rare case and there are diagnostic enigmas in the field we think that complete neuroimaging data and intraoperative views are necessary, so we apologize that the number of the figures are 5. including th
It would be a great honor if you accept this paper in your respectful journal.

Sincerely,

Dr. Amir Saied Seddighi  
Dr. Afsoun Seddighi